
Daily Report for Oct 11 2013
IOP #2 begins tonight with a balloon launch at 5 pm LST.

Weather is cool and crisp with clear skies.

Inspection of 449 site went well. Rass speakers all working, and fan on tower is working as well.

Profiler display shows no problems, Spectral profiles are all in agreement and dwell counter is working.

Nagios screen shows red flags for lidar, however earlier entry by Tim indicates system is up and running. Both profiler_anyfail_prof and 
profiler_anyfail_surface are flagged as red as well. Same problem occurred yesterday and problem seemed to fix itself after some time.  I will keep an eye 
on it.

Surface met data looks good.

Daily amp power is at .139 mW
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